Author's response to reviews

Title: Regional-scale climate-variability synchrony of cholera epidemics in West Africa.

Authors:

Guillaume Constantin de Magny (g.demagny@gmail.com)
Jean-Francois Guegan (jf.guegan@mpl.ird.fr)
Michel Petit (michel.petit@ird.fr)
Bernard Cazelles (cazesles@biologie.ens.fr)

Version: 3 Date: 16 March 2007

Author's response to reviews: see over
Montpellier, March 16, 2007

The Editor in Chief,

BMC Infectious Diseases

Dear Editor in Chief,

Please find the final revision of the manuscript entitled “Regional-scale climate-variability synchrony of cholera epidemics in West Africa” by G. Constantin de Magny, J.-F. Guégan, M. Petit, and B. Cazelles for publication to your journal.

As requested by the BioMed Central Editorial Production Team, the article integrates all the technical comments:
- 1: MATERIALS AND METHODS -> Methods
- 2: remove footnotes, use references
- 3: Website addresses as reference in the following structure, e.g. 17. Website title [www.website.com].
- 4: tables are now referenced in a numerically consecutive fashion.
- 5: Table title and legend changed.
- 6: Figures -> Figure legends

The authors hope that the article is suitable for publication in the journal BMC Infectious Diseases.

Yours sincerely,

Guillaume Constantin de Magny
G.E.M.I. UMR CNRS-IRD 2724 and US ESPACE
E- mail: g.demagny@gmail.com